Author's response to reviews

Title: Effects of acute ingestion of different fats on oxidative stress and inflammation in overweight and obese adults.

Authors:

Abigail D Peairs (abigail.peairs@uc.edu)
Janet W Rankin (jrankin@vt.edu)
Yong Woo Lee (ywlee@vt.edu)

Version: 4 Date: 3 November 2011

Author's response to reviews: see over
November 3, 2011

Dr. Nehme Gabriel  
Editor in Chief of *Nutrition Journal*

Dear Dr. Gabriel,

We are pleased that our manuscript has been found acceptable for publication in the *Nutrition Journal*. We have made the final adjustments requested as outlined below. Please do not hesitate to contact me if you have any questions.

Sincerely,

Abigail Peairs, PhD  
Assistant Professor  
Department of Nutritional Sciences  
University of Cincinnati  
Abigail.peairs@uc.edu

**Final corrections made:**  
1. Author list adjusted to include only first names, middle initial, and last name (Dr. Lee’s first name is “Yong Woo”).

2. We removed the running title and counts from the title page.

3. We removed the degrees listed from the author list.

4. We added “USA” as the country affiliation for all authors.

5. We removed the highlighting that had been added for revisions.

6. We moved the competing interests section to after the abbreviations.

7. We moved the figure legends and tables to after the reference list.

8. We remove the figure titles from the powerpoint figure images.

9. The manuscript was checked for typographical errors, the most commonly changed item was to change “k”s in NF-kB to make the correct abbreviation with the kappa sign (κ) as it should be: NF-κB. (pages abstract, 10, 11, 13)

We also had to add the “1” to ICAM in several places to make it “ICAM-1.” (pages 10, 12, 13).